Author’s response to reviews

Title: Postoperative chemoradiotherapy is superior to postoperative chemotherapy alone in squamous cell lung cancer patients with limited N2 lymph node metastasis

Authors:

Liyu Su (suliyu199204@fjmu.edu.cn)
Mingqiu Chen (drchenmingqiu@163.com)
Huiyan Su (691382659@qq.com)
Yaqing Dai (1729194154@qq.com)
Shaoxing Chen (853136809@qq.com)
Jiancheng Li (jianchengli_jack@126.com)

Version: 4 Date: 30 Aug 2019

Author’s response to reviews:

Dear Editors and Reviewers:

Thank you very much for your email and comments concerning our manuscript entitled “Postoperative chemoradiotherapy is superior to postoperative chemotherapy alone in squamous cell lung cancer patients with limited N2 lymph node metastasis” (ID: BCAN-D-19-00741R3).

We reviewed all comments carefully and revised the manuscript according to the reviewers` comments. The followings are the responds to the reviewer`s comments.

Editor

1. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

Response: Thank you very much for your suggestion! We revised them in the revised manuscript.
2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Thank you very much for such suggestion!

Thank you very much for your comments and suggestion!